Author’s response to reviews

Title: The influence of probiotics on genital high-risk human papilloma virus clearance, associated lesions and quality of cervical smear: A randomized placebo-controlled trial

Authors:

Yu-Che Ou (tedycou@gmail.com)
Hung-Chun Fu (allen133@adm.cgmh.org.tw)
Chih-Wen Tseng (otcwwell@adm.cgmh.org.tw)
Chen-Hsuan Wu (chenhsuan5@gmail.com)
Ching-Chou Tsai (nick@adm.cgmh.org.tw)
Hao Lin (haolin423700@gmail.com)

Version: 3 Date: 02 Jul 2019

Author’s response to reviews:

Dear Editor Prof. Tovah Honor Aronin,

We thank to the Editor for further valuable comments. The material presented in manuscript has been revised according to the editor’s recommendation. All authors agree with the content of the article and approve of its re-submission to the journal. Thank you for giving us a chance to publish our article in this journal.

Again, on behalf of all authors, we thank the reviewers for their valuable comments, which led to significant improvements to the manuscript.

Respectfully,

Hao Lin,

Department of Obstetrics and Gynecology, Kaohsiung Chang Gung Memorial Hospital, Taiwan, R.O.C.